Author’s response to reviews

Title: Successful Treatment of Gemcitabine-Induced Acute Interstitial Pneumonia With Imatinib Mesylate: a Case Report

Authors:

Elisabetta Fenocchio (elisabetta.fenocchio@ircc.it)
Ilaria Depetris (ilaria.depetris@ircc.it)
Delia Campanella (delia.campanella@ircc.it)
Lucia Garetto (lucia.garetto@ircc.it)
Fabrizio Carnevale Schianca (fabrizio.carnevale@ircc.it)
Danilo Galizia (danilo.galizia@ircc.it)
Giovanni Grignani (giovanni.grignani@ircc.it)
Massimo Aglietta (massimo.aglietta@ircc.it)
Francesco Leone (francesco.leone@ircc.it)

Version: 1 Date: 06 Jul 2016

Author’s response to reviews:

Dear Editor,

Please find enclosed the manuscript entitled “Successful treatment of gemcitabine-induced acute interstitial pneumonitis with imatinib mesylate”, with the changes suggested by the reviewers.

Being a single case experience, we agreed with the suggestion to weaken the conclusions on imatinib's efficacy in treating idiopathic fibrosis.

We have included a short overview on the latest treatment options for idiopathic fibrosis as requested.

Finally, we confirm that this work is original and has not been published or submitted to another scientific journal and it is not currently under consideration for publication elsewhere.
Please address all correspondence concerning this manuscript to me at francesco.leone@irc.c.it.

Thank you for your consideration of this manuscript.

Sincerely,

Francesco Leone